Author’s response to reviews

Title: Computer-associated health complaints and sources of ergonomic instructions in computer-related issues among Finnish adolescents: A cross-sectional study

Authors:

Paula T. Hakala (paula.t.hakala@uta.fi)
Lea A. Saarni (lea.saarni@uta.fi)
Ritva L. Ketola (ritva.ketola@ttl.fi)
Erja T. Rahkola (erja.rahkola@ramk.fi)
Jouko J. Salminen (jouko.salminen@tyks.fi)
Arja H. Rimpelä (arja.rimpela@uta.fi)

Version: 8 Date: 4 December 2009

Author’s response to reviews:

We have revised the article taking the Editorial Production Team’s comment into account. The following correction has been made:
1. Tables have been new created.
2. Abstract in the submission system and in the manuscript are the same now.